Author's response to reviews

Title: Low risk of attrition among adults on antiretroviral therapy in the Rwandan national program: A retrospective cohort analysis of 6, 12, and 18 month outcomes.

Authors:

Harriet Nuwagaba-Biribonwoha (hn2158@columbia.edu)
Aleksandra Jakubowski (a ej2121@mail.cumc.columbia.edu)
Veronicah Mugisha (vm2208@columbia.edu)
Paulin Basinga ( paulin.basinga@gatesfoundation.org)
Anita Asiimwe (anita.asiimwe@gmail.com)
Batya Elul, Elul (be2124@mail.cumc.columbia.edu)

Version: 6
Date: 21 July 2014

Author's response to reviews:

Dear Editors,

The following editorial changes have been made:
(a) Deleted the Funding section.
(b) Added the funding information to the Acknowledgements section.

Thank you.
The Authors